WILEY

Janssen-Sponsored Satellite Symposium at the 30th EADV Virtual Congress 2021



The art of joint forces: crafting psoriatic arthritis care for dermatologists

This virtual satellite symposium will focus on the necessity for practicing dermatologists to understand the burden of psoriatic arthritis in patients with psoriasis. It will emphasize how important it is that dermatologists detect early signals of psoriatic arthritis in patients with psoriasis and also understand why targeting IL-23 directly can be effective in treating and potentially also preventing the development of psoriatic arthritis for their psoriasis patients



Check for updates

Incidence, prevalence and mortality of bullous pemphigoid in England 1998–2017: a population-based cohort study*

M.S.M. Persson $(\mathbf{D}, \mathbf{1}, \mathbf{K}, \mathbf{E}, \mathbf{H}, \mathbf{T}, \mathbf{V})$ Vinogradova,² S.M. Langan,³ J. Hippisley-Cox,⁴ K.S. Thomas $(\mathbf{D}, \mathbf{1}, \mathbf{T})$ and S. Gran $(\mathbf{D}, \mathbf{1}, \mathbf{V})$

¹Centre of Evidence Based Dermatology, University of Nottingham, Nottingham, UK

²Division of Primary Care, University of Nottingham, Nottingham, UK

³Department of Non-communicable Disease Epidemiology, London School of Hygiene and Tropical Medicine, London, UK

⁴Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

Linked Comment: Bax and Werth. Br J Dermatol 2021; 184:5-6.

Summary

Correspondence

M.S.M. Persson Email: Monica.Persson1@nottingham.ac.uk

Accepted for publication 5 March 2020

Funding sources

This paper presents independent research funded by the National Institute for Health Research (NIHR) under its Research for Patient Benefit Programme (Grant Reference Number PB-PG-0817-20033). The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. S.M.L. reports grants from Wellcome Senior Clinical Fellowship in Science (205039/Z/16/Z) during the conduct of the study.

Conflicts of interest

All authors declare no conflicts of interest.

*Plain language summary available online

DOI 10.1111/bjd.19022

Background A rising incidence and high mortality were found for bullous pemphigoid (BP) over a decade ago in the UK. Updated estimates of its epidemiology are required to understand the healthcare needs of an ageing population.

Objectives To determine the incidence, prevalence and mortality rates of BP in England from 1998 to 2017.

Methods We conducted a cohort study of longitudinal electronic health records using the Clinical Practice Research Datalink and linked Hospital Episode Statistics. Incidence was calculated per 100 000 person-years and annual point prevalence per 100 000 people. Multivariate analysis was used to determine incidence rate ratios by sociodemographic factors. Mortality was examined in an age-, sex- and practice-matched cohort, using linked Office of National Statistics death records. Hazard ratios (HRs) were stratified by matched set.

Results The incidence was 7.63 [95% confidence interval (CI) 7.35-7.93] per 100 000 person-years and rose with increasing age, particularly for elderly men. The annual increase in incidence was 0.9% (95% CI 0.2-1.7). The prevalence almost doubled over the observation period, reaching 47.99 (95% CI 43.09-53.46) per 100 000 people and 141.24 (95% CI 125.55-158.87) per 100 000 people over the age of 60 years. The risk of all-cause mortality was highest in the 2 years after diagnosis (HR 2.96; 95% CI 2.68-3.26) and remained raised thereafter (HR 1.54; 95% CI 1.36-1.74).

Conclusions We report a modest increase in the incidence rate of BP, but show that the burden of disease in the elderly population is considerable. Mortality is high, particularly in the first 2 years after diagnosis.

What is already known about this topic?

- Bullous pemphigoid (BP) is a blistering skin disorder which typically affects the elderly.
- BP poses a high burden on affected patients and has significant healthcare costs.
- The burden of disease in the UK was estimated over a decade ago and found to have a rising incidence and high mortality.

What does this study add?

- The incidence of BP in England was 7.63 (95% CI 7.35-7.93) per 100 000 person-years between 1998 and 2017.
- The burden of BP in the elderly population is substantial and it should not be considered a rare disease in these age groups.

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists. This is an open access article under the terms of the Creative Commons Attribution License, which permits use,

distribution and reproduction in any medium, provided the original work is properly cited.

The risk of death was almost three times higher in the first 2 years after a diagnosis of BP compared with people without the disease and remained raised thereafter.

Bullous pemphigoid (BP) is a rare autoimmune disease, characterized by blistering of the skin. It is predominantly a disease of the elderly, and is associated with high morbidity, mortality and significant healthcare costs.^{1,2} A previous study of BP in the UK estimated that the incidence was increasing by 17% per annum between 1996 and 2006.³ However, this work was conducted over a decade ago and included only patients identified in primary care. More severe cases whose diagnosis was not returned to their general practitioner (GP), potentially because they died in hospital shortly after being diagnosed, would not have been captured. An updated estimate is required to quantify the burden of disease in England, which may be an important consideration when assessing the healthcare needs of an ageing population.

Routinely collected electronic healthcare records are valuable for examining a rare disease like BP, because of the availability of data on a large number of patients that are broadly representative of the UK population. Patients with BP can be identified using a validated method and code list with a positive predictive value of 93.2% [95% confidence interval (CI) 91.3-94.8).⁴ The incidence and prevalence rates can be determined with high precision, while longitudinal follow-up allows long-term outcomes, such as death, to be examined. The aim of the present work was to determine the incidence, prevalence and mortality of BP in England between 1998 and 2017.

Patients and methods

Study design

Incidence and prevalence were examined using a retrospective cohort. An age-, sex- and practice-matched longitudinal cohort was used to compare the mortality of BP with the general population. We followed RECORD guidelines for the reporting of studies conducted using observational routinely collected health data.⁵

Data sources

Clinical Practice Research Datalink

The CPRD GOLD is a longitudinal database of UK general practices using the Vision software system. It contains anonymized healthcare records for approximately 17 million patients, with a current coverage of approximately 2.7 million (4%) of the UK population. Routinely collected clinical data generated from consultations, hospital discharges or specialist clinic letters are recorded using Read codes.⁶ Diagnoses of BP are predominantly made in secondary care, but are subsequently returned to GPs and entered into primary care records.

The data in the GPRD have repeatedly been shown to be of good research quality.⁷ At practice level, participating practices are assigned an 'up-to-standard' date on completion of regular audits confirming data quality. At the patient level, records are assessed and patients are deemed 'acceptable' if data checks indicate that their record meets prespecified quality standards.

Linked data

Approximately 75% of English practices, covering more than 10 million patients, have consented to provide patient-level information from linked resources.⁸ The individual-level linked sources used in this study were Hospital Episode Statistics (HES) admitted patient care data and Office for National Statistics (ONS) death registration data. HES admitted patient care contains diagnoses for each hospitalization episode, coded using the International Classification of Diseases version 10 (ICD-10).^{9,10} Death registration data contain the official date of death.

Study population

Adult men and women registered with 410 HES-linked general practices in England during the period of 1 January 1998 to 31 December 2017 were included. Patients were eligible from the date their practice was deemed to be contributing up-to-standard data and their record was verified as 'acceptable' research quality.

Outcomes

For incidence and prevalence, the outcome was a diagnosis of BP in the CPRD or HES. The date of death was the outcome for the all-cause mortality analyses.

Case definition of bullous pemphigoid

A diagnosis of BP is usually made on referral to a dermatology department, where patients undergo clinical examination and laboratory investigations, usually a skin biopsy for histology and direct immunofluorescence. The diagnosis is consequently communicated to primary care, where it is entered in the CPRD using Read codes, or to the person's inpatient records, where it is entered in the HES admitted patient care records using ICD-10 codes. People with BP were adult patients with a diagnostic code for BP in the CPRD or HES. An algorithm was applied to the CPRD to identify patients with Read codes for BP (M145, 'bullous pemphigoid'; M145·00, 'pemphigoid'; or M145z00, 'pemphigoid not otherwise specified').⁴ A similar algorithm, using ICD-10 diagnostic codes, was applied to HES admitted patient care data (L12·0, 'bullous pemphigoid'; or L12·9, 'pemphigoid, unspecified'). The index date of diagnosis for people with BP was the first date of occurrence of the Read or ICD-10 code generated from the algorithm.

Incident diagnoses were those with an index date at least 1 year after their current registration date with their GP. The 1-year lag period was imposed to minimize the risk of prevalent cases being identified as incident ones.¹¹ Prevalent diagnoses were those with a record of BP at any point before the end of the observation period.

Matched disease-free individuals

Up to four disease-free unexposed individuals were selected randomly for each case and matched by age (within 1 year of date of birth), sex and general practice. They were assigned an index date, which was the date of diagnosis of BP of the case they were matched to. The matched disease-free individuals had to be alive and contributing data at the time of the index date.

Observation period

The observation period commenced on the latest of (i) 1 January 1998, (ii) the date the patient registered with their current practice, (iii) the practice's up-to-standard date, or (iv) the patient's 18th birthday.

The observation period terminated on the earliest of (i) 31 December 2017, (ii) the date of death, (iii) the date the patient left the practice, (iv) the practice's last data collection date, or (v) the most recent linkage date between CPRD and HES.

Statistical analysis

Incidence

The incidence rates per 100 000 person-years overall and by age (10-year bands), calendar period (3-year bands), sex, ethnicity, region and index of multiple deprivation were calculated. Poisson regression was used to determine incidence rate ratios (RRs) for each sociodemographic factor. For the unordered categorical factors, region and ethnicity, the group with the most person-years of follow-up was chosen as the reference category. Multivariate regression was used to adjust for age, sex and calendar period. A post hoc examination of the interaction between age and sex, adjusted for calendar period, was conducted.

The crude and age- and sex-adjusted annual changes in incidence were determined over the 20-year observation period. The crude incidence rate for each calendar year, with direct standardization to the European Standard Population of 2013, was determined.

Annual point prevalence

The number of individuals with BP contributing data on 31 July of each calendar year was divided by the total number of adults alive and contributing data at the same point in time.

Mortality

People with BP and their matched disease-free individuals were followed from their index date and all deaths in the cohort were identified. Crude mortality rates were calculated per 1000 person-years. Kaplan–Meier techniques were used to determine 1-year mortality and median survival time. All-cause mortality in people with vs. without BP was compared using Cox proportional hazards regression stratified by matched set to account for age, sex, general practice and calendar period. The proportional hazards assumption, tested using Schoenfeld's residual, was violated. The observation period was therefore divided into two follow-up periods: within 2 years and 2 years or more after the index date.

Study size

This was a population-based cohort study that used all available data. Assuming only one matched disease-free individual was found per patient with BP, this study would have > 90% power to detect a HR of $2 \cdot 1$ or greater at the 5% significance level.³

Sensitivity analyses

The incidence, prevalence and mortality of BP in the UK were established using all primary care data available in the CPRD, irrespective of linkages. The results were compared with the primary analyses, which included only patients registered with HES-linked practices. Incidence RRs for ethnicity were determined only after 2006 as ethnicity data were more poorly recorded for people who registered with the CPRD prior to this.¹²

Unless otherwise specified, the P-values presented were obtained from the likelihood ratio test and P < 0.05 was considered statistically significant. Analyses were conducted with Stata 16 (2019; StataCorp LLC, College Station, TX, USA).

Ethical approval

The present study was approved by the Independent Scientific Advisory Committee for the CPRD (ISAC protocol no 18_224).

Results

Study population

There were 2658 incident individuals with BP identified during the observation period: 2468 from English HES-linked general practices in the CPRD (Figure 1) and an additional 190 individuals with a diagnosis of BP only in their HES records. Median age at presentation was 81 (interquartile range 72–87) years and 1497 (56·3%) of patients were women.

Incidence

The pooled incidence rate over the 20-year period was 7·63 (95% CI 7·35–7·93) per 100 000 person-years (Table 1). The crude incidence rate was 25% higher in women (RR 1·25, 95% CI 1·16–1·35), but after adjustment for age and calendar period, the incidence rates were 13% lower in women (RR 0·87, 95% CI 0·80–0·94) than men. Incidence rates rose dramatically with age, such that the highest rates were recorded in patients over 90 years. The increase in incidence in older age groups was more marked for men than women (P-value for interaction < 0·001, Figure 2). After adjustment for age, sex and calendar period, rates were highest in those of Asian ethnicity and lowest in those of unknown ethnicity compared to whites, but ethnicity data were missing in 26% of the whole study population. Incidence varied by geographic region, but not by level of deprivation.

Between 1998 and 2017, incidence increased by 1·2% (RR 1·012; 95% CI 1·005–1·020; P = 0·002) per calendar year. After adjustment for age and sex, the estimated annual increase in incidence was 0·9% (RR 1·009; 95% CI 1·002–1·017; P = 0·016). Overall increase in incidence over the 20-year study period (2017 vs. 1998) was 16·4% after adjustment for age and sex (RR 1·164; 95% CI 0·805–1·684; P = 0·420). Annual incidence rates, calculated by direct standardization to the European Standard Population, are illustrated in Figure S1 (see Supporting Information).

Prevalence

The prevalence of BP was 26.82 (95% CI 23.83-30.19) per 100 000 people in 1998 and increased to 47.99 (95% CI 43.09-53.46) per 100 000 in 2017 (Figure 3). The most recent estimate of the prevalence of BP in those aged over 60 years was 141.24 (95% CI 125.55-158.87) per 100 000 people in 2017. In the over-80s, the prevalence was 375.02 (95% CI 320.31-439.04) per 100 000.

Mortality

The cohort included 2639 people with BP and 10 463 matched disease-free individuals (details provided in Supporting Information; Text S1). Within the cohort 1237 deaths occurred in the individuals with BP and 3427 in those without, equating to a crude mortality rate (per 1000 personyears) of 138.82 (95% CI 131.30–146.78) for people with BP and 74.56 (95% CI 72.11–77.10) for people without (Table 2). The median survival time was 4.93 years for people with BP and 9.27 years for people without (Figure 4). The 1year mortality for BP was 20.36% (95% CI 18.82–22.00) and 7.03% (95% CI 6.54–7.55) for disease-free individuals.

After stratifying by matched set to account for age, sex and general practice, the HR was 2.96 (95% CI 2.68-3.26) in the first 2 years and 1.54 (95% CI 1.36-1.74) thereafter (Table 2).

Sensitivity analyses

All results were confirmed in sensitivity analyses using patients from all of the CPRD compared with the main analysis, which examined HES-linked practices only (Texts S2–S4, Figures S2, S3, Tables S1, S2; see Supporting Information).

Discussion

Our main findings are as follows. The incidence of BP during the 20-year observation period was 7.63 (95% CI 7.35–7.93) per 100 000 person-years. The incidence was highest in the elderly, particularly in elderly men. Between 1998 and 2017, the incidence of BP increased by 0.9% (95% CI 0.2–1.7) per calendar year and the prevalence almost doubled from 26.82 (95% CI 23.83–30.19) per 100 000 people in 1998 to 47.99 (95% CI 43.09–53.46) per 100 000 people in 2017. Using the European definition of a rare disease (prevalence < 50 per 100 000 people),¹³ we have for the first time shown that BP is not a rare disease in the elderly (\geq 60 years).

The risk of death within 2 years of diagnosis was almost three times higher in patients with BP than in matched disease-free individuals (HR 2.96, 95% CI 2.68–3.26). The mortality remained raised throughout the remainder of the observation period (HR 1.54, 95% CI 1.36–1.74).

There are comparisons that can be made with other studies. An increasing incidence of BP has previously been reported in the UK using data from The Health Improvement Network (THIN) up to 2006.³ The current data show a more modest increase in incidence over time, but a higher incidence rate across the whole observation period. The reasons for these differences are likely multifactorial, although largely driven by differences between the databases used. Significantly lower incidence rates have been reported for venous leg ulcers in THIN compared with the General Practice Research Database (the predecessor of CPRD) using data collected prior to 2000.¹⁴ The low incidence rate of BP observed using THIN, particularly prior to 1998, may be artefactual and could explain why a more dramatic increase in incidence over time was found. Other factors that may explain the discrepancy include small differences in either the numerator or denominator between the present and the earlier study. For example, the present study used an algorithm to identify patients and included patients that had records of BP in HES, but not

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists.



Figure 1 Algorithm used to identify incident adult individuals with bullous pemphigoid in the Clinical Practice Research Datalink (CPRD) between January 1998 and December 2017.

CPRD. Nevertheless, the absolute difference in incidence between the studies remains small and reduces over time.

Our incidence results are similar to those reported recently in Sweden.¹⁵ Using the National Patient Register the authors reported an incidence of 7.1 (95% CI 6.5–7.7) per 100 000 person-years.¹⁵ In contrast, studies conducted elsewhere in the world report lower results (Table S3; see Supporting Information).^{3,15–28} This may be due to underlying differences in risk between populations or because of differences in methods. For example, other studies generally present rates for the population including children, who are rarely affected. In addition, other studies are generally smaller, hospital based and may not achieve complete capture of all cases within the defined denominator as patients managed in primary care or referred to other dermatology centres outside the geographical boundaries of the study are often missed.

The 1-year mortality of 20.36% in the present study is similar to previous estimates from the UK. Between 1996 and 2006, the 1-year mortality in the UK using THIN was 19% (95% CI 16.2-21.8).³ Between 1991 and 2001 the 1-year mortality of the 83 incident cases identified from the Pathology and Immunology database of a single referral centre in

						Mutually adjusted	
Variable	Person-years	Events	Incidence rate (CI)	Crude RR (CI)	P-value	incidence RRs (CI)	P-value
Pooled	34 825 210	2658	7.63 (7.35–7.93)	-		-	
Sex					< 0.001		< 0.001
Male	17 171 950	1161	6.76 (6.38–7.16)	1.00		1.00	
Female	17 653 250	1497	8.48 (8.06-8.92)	1.25 (1.16–1.35)		0.87 (0.80 - 0.94)	
Age group (years)					$< 0.001^{a}$		< 0.001
< 60	24 467 650	254	1.04 (0.92–1.17)	1.00		1.00	
60–69	4 765 380	264	5.54 (4.91-6.25)	5.34 (4.49–6.34)		5.34 (4.49–6.34)	
70–79	3 423 240	662	19.34 (17.92–20.87)	18.63 (16.12-21.53)		18.79 (16.26–21.72)	
80-89	1 812 280	1074	59.26 (55.82-62.92)	57.09 (49.79–65.45)		58.12 (50.67–66.67)	
≥ 90	356 660	404	113.27 (102.75–124.88)	109.12 (93.27-127.66)		112.88 (96.36–132.23)	
Calendar year(years)					0.007^{a}		0.049
1998-2000	3 702 250	247	6.67 (5.89–7.56)	1.00		1.00	
2001-2003	5 543 890	405	7.31 (6.63–8.05)	1.10 (0.94–1.28)		1.07 (0.91–1.26)	
2004-2006	6 288 600	470	7.47 (6.83–8.18)	1.12 (0.96–1.31)		1.09 (0.94–1.28)	
2007-2009	6 605 740	508	7.69 (7.05–8.39)	1.15 (0.99–1.34)		1.12 (0.96–1.30)	
2010-2012	6 369 500	534	8.38 (7.70–9.13)	1.26 (1.08–1.46)		1.20 (1.03–1.40)	
2013-2015	4 745 980	363	7.65 (6.90-8.48)	1.15 (0.98–1.35)		1.08 (0.92–1.27)	
2016-2017	1 569 240	131	8.35 (7.03–9.91)	1.25 (1.01–1.55)		1.21 (0.98–1.50)	
Ethnicity					< 0.001		< 0.00
White	23 950 960	2318	9.68 (9.29–10.08)	1.00		1.00	
Black ^b	484 490	14	2.89 (1.71–4.88)	0.30 (0.18-0.51)	< 0.001	0.80 (0.48 - 1.36)	0.415
Asian ^b	965 200	47	4.87 (3.66–6.48)	0.50 (0.38–0.67)	< 0.001	1.37 (1.03–1.84)	0.03
Other ^b	418 380	23	5.50 (3.65-8.27)	0.57 (0.38–0.86)	0.007	1.22 (0.81–1.84)	0.342
Unknown ^b	9 006 180	256	2.84 (2.52-3.21)	0.29 (0.26-0.33)	< 0.001	0.63 (0.55–0.72)	< 0.00
Region					< 0.001		0.02
North West	5 374 400	359	6.68 (6.02-7.41)	1.00		1.00	
North East	799 330	43	5.38 (3.99–7.25)	0.81 (0.59–1.11)		0.81 (0.59–1.11)	
Yorkshire &	1 402 950	100	7.13 (5.86-8.67)	1.07 (0.86–1.33)		1.02 (0.82–1.28)	
The Humber							
East Midlands	1 052 510	65	6.18 (4.84–7.88)	0.93 (0.71–1.20)		0.95 (0.73-1.24)	
West Midlands	4 254 880	307	7.22 (6.45-8.07)	1.08 (0.93–1.26)		0.99 (0.85-1.15)	
East of England	4 007 770	316	7.89(7.06 - 8.80)	1.18 (1.02–1.37)		1.12 (0.96–1.30)	
South West	4 355 830	430	9.87 (8.98–10.85)	1.48 (1.29–1.70)		1.20 (1.04-1.38)	
South Central	4 543 990	377	8.30 (7.50-9.18)	1.24 (1.08–1.44)		1.14 (0.99–1.32)	
London	4 352 770	308	7.08 (6.33-7.91)	1.06 (0.91–1.23)		1.15 (0.99–1.34)	
South East Coast	4 680 780	353	7.54 (6.79–8.37)	1.13 (0.98–1.31)		0.99 (0.85-1.14)	
Deprivation					0.005*		0.208
1	8 312 070	657	7.90 (7.32-8.53)	1.00		1.00	
2	7 702 910	601	7.80 (7.20-8.45)	0.99 (0.88-1.10)		0.94 (0.84-1.05)	
3	7 260 820	610	8.40 (7.76-9.10)	1.06 (0.95–1.19)		1.02 (0.91–1.14)	
4	6 272 660	438	6.98 (6.36-7.67)	0.88 (0.78–1.00)		0.91 (0.80-1.02)	
5	5 276 750	352	6.67 (6.01-7.41)	0.84 (0.74-0.96)		0.93 (0.81-1.06)	

Table 1 Univariate and multivariate analysis of incidence of bullous pemphigoid in Hospital Episode Statistics-linked general practices and hospital inpatient records in England per 100 000 person-years

^aLR test for trend; ^bWald's test; all other P-values are from LR test. Mutually adjusted model is adjusted for age, sex and calendar year. CI, confidence interval; HES, Hospital Episode Statistics; LR, likelihood-ratio test; RR, rate ratio

North-east Scotland was 25%.²³ A recent meta-analysis estimated the pooled 1-year mortality rate in European populations to be 26.7% (95% CI 22.2–31.2), although the heterogeneity between studies was large.²⁸ Survival in the first year after diagnosis appears marginally better in England than in the rest of Europe, but worse than the USA (15.1%, 95% CI 7.9–22.3). Such differences in mortality could be explained by differences between populations, methods of case and outcome ascertainment, temporal trends, or true differences in the mortality between nations.

The strengths of this study include a large sample size, a population that is representative of the UK general population, use of a validated approach to identify cases, accurate recording of death dates, and using linked primary and secondary care data to get a more comprehensive coverage of cases. Using routinely recorded data on a population level makes this study less susceptible to selection bias.

Case ascertainment is a limitation as it is reliant on the presence of clinical codes indicative of BP. Inaccurate recording may have affected the estimates. We know from previous

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists.



Figure 2 Crude incidence rate of bullous pemphigoid in English Hospital Episode Statistics-linked general practices and hospital inpatient records per 100 000 person-years per age category. Rates shown separately for men and women, with upper and lower limits of 95% confidence intervals.



Figure 3 Point prevalence of bullous pemphigoid on 31 July of each calendar year in English Hospital Episode Statistics-linked general practices, per 100 000 people. Prevalence rates shown for all adults and for people aged over 60 years, with shading indicating the upper and lower limits of the 95% CI.

work that a small proportion of those identified with BP codes are likely to have other diagnoses, such as mucous membrane pemphigoid.⁴ Inversely, some patients with BP may be missed as they are coded as having pemphigus vulgaris. However, the positive predictive value of the BP codes in this study is 93%, supporting our conclusions. Additionally, it is difficult to interpret findings related to ethnicity due to poor recording in those registered prior to 2006. Finally, our work has assumed that once a patient has developed BP they remain a prevalent case until the end of follow-up. In reality, patients may enter remission or become

disease-free, and the true prevalence of BP may be more conservative than that presented.

The possible explanations for our findings follow here. The incidence of BP rises with increasing age, which may reflect an aetiology related to the biological process of ageing. This effect was more marked in men. An alternative explanation may be the age-related rise in prevalence of diseases associated with the development of BP, such as dementia and Parkinson's disease.^{30,31} Alternatively, the relationship may reflect increasing use of drugs that induce BP in the elderly population. Variation in the incidence of BP was also observed between

Table 2 Hazard ratios (HRs) for death, stratified by matched set, comparing patients with bullous pemphigoid and matched disease-free people from Hospital Episode Statistics-linked general practices and hospital inpatient records in England. Mortality rates are presented per 1000 personyears

Variable	Person-years	Events	Mortality rate (95% CI)	Stratified HR ^a (95% CI)	P-value
Pooled ^b					
Controls	45 961.5	3427	74.56 (72.11–77.10)	_	
Cases	8910.7	1237	138.82 (131.30-146.78)		
< 2 years					
Controls	17 374.5	1270	73.10 (69.18-77.23)	1.00	< 0.001
Cases	3759.0	734	195.26 (181.64-209.91)	2.96 (2.68–3.26)	
\geq 2 years					
Controls	28 578.0	2157	75.45 (72.34-78.71)	1.00	< 0.001
Cases	5151.7	503	97.64 (89.47-106.55)	1.54 (1.36–1.74)	

^aHR stratified by matched set to account for age, sex, general practice and calendar period; ^bstratified HR not calculated for the pooled follow-up period as the proportional hazards assumption was violated. CI, confidence interval



Figure 4 Kaplan-Meier curve for patients with and without bullous pemphigoid, showing mortality over time since index date and number of patients remaining in the study.

ethnic groups and geographic regions. The antidiabetic drug dipeptidyl peptidase-4 inhibitor (DPP-4i) is associated with an approximate doubling of the risk of BP^{32–36} and the increased risk in Asians, compared with the white ethnic group, may partly relate to the increased levels of diabetes in this ethnic group. The rising trend for prescriptions of DPP-4i for diabetes throughout the study period may also contribute to the increased incidence observed.^{37,38} Regional differences could point to environmental factors, perhaps to differences in

incidence between urban and rural communities,¹⁸ or could be due to differences in the prescribing patterns or awareness of BP in the medical communities. The rising prevalence of BP may be a result of the increasing incidence of the disease or may reflect that the English population is living longer.³⁹ Alternatively, it may be that the prevalence nearer the date of database inception may be underestimated. People with diagnoses prior to the inception of CPRD may not have had the disease entered retrospectively into their electronic records.

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists.

In terms of possible clinical implications of our findings, healthcare professionals should be made aware that BP is not a rare disease in the elderly. A greater awareness of the disease will hopefully reduce diagnostic delay. The mortality risk from BP has remained largely unchanged since it was last examined over a decade ago. Improved management, a better understanding of the aetiology, and examination of the causes of death in BP are still required.

In conclusion, BP is a rising concern for the ageing population of England. In the elderly population, it should no longer be considered a rare disease. Its incidence increases with age and it is associated with almost a three times greater chance of death in the first 2 years after diagnosis compared with disease-free individuals. The increased burden of disease, particularly in the elderly, requires earlier diagnosis, improved treatment and better understanding of the aetiology.

Acknowledgments

This study is based in part on data from the Clinical Practice Research Datalink obtained under licence from the UK Medicines and Healthcare Products Regulatory Agency. The data is provided by patients and collected by the National Health Service as part of their care and support. Hospital Episode Statistics and Office of National Statistics data, 2019[©], is re-used with the permission of The Health & Social Care Information Centre. All rights reserved. The interpretation and conclusions contained in this study are those of the authors alone.

References

- 1 Ständer S, Färber B, Radeke S et al. Assessment of the healthcare costs for patients with pemphigus and bullous pemphigoid in an academic centre in Germany. Br J Dermatol 2020; doi: 10.1111/bjd.18731.
- 2 Kouris A, Platsidaki E, Christodoulou C et al. Quality of life, depression, anxiety and loneliness in patients with bullous pemphigoid. A case control study. An Bras Dermatol 2016; 91:601–3.
- 3 Langan S, Smeeth L, Hubbard R et al. Bullous pemphigoid and pemphigus vulgaris incidence and mortality in the UK: population based cohort study. BMJ 2008; **337**:a180.
- 4 Persson MSM, Harman KE, Vinogradova et al. Incidence, prevalence, and mortality of bullous pemphigoid in England 1998–2017: a population-based cohort study. Br J Dermatol 2020; doi: 10.1111/bjd.19022.
- 5 Benchimol EI, Smeeth L, Guttmann A et al.; RECORD Working Committee. The REporting of studies Conducted using Observational Routinely-collected health Data (RECORD) statement. PLoS Med 2015; 12:e1001885.
- 6 Chisholm J. The Read clinical classification. BMJ 1990; 300:1092.
- 7 Herrett E, Thomas SL, Schoonen WM et al. Validation and validity of diagnoses in the General Practice Research Database: a systematic review. Br J Clin Pharmacol 2010; 69:4–14.
- 8 Clinical Practice Research Datalink (CPRD). Release Notes: CPRD GOLD. June 2019. London: Medicines and Healthcare Products Regulatory Agency, 2019. Available at: https://www.cprd.com/.
- 9 World Health Organization. International Statistical Classification of Diseases and Related Health Problems: 10th revision (ICD-10), 1992 version. Geneva: WHO. Available at: http://www.who.int/classifications/apps/icd/icd.

- 10 Herbert A, Wijlaars L, Zylbersztejn A et al. Data resource profile: Hospital Episode Statistics Admitted Patient Care (HES APC). Int J Epidemiol 2017; 46:1093–1093i.
- 11 Lewis JD, Bilker WB, Weinstein RB, Strom BL. The relationship between time since registration and measured incidence rates in the General Practice Research Database. Pharmacoepidemiol Drug Saf 2005; 14:443–51.
- 12 Mathur R, Bhaskaran K, Chaturvedi N et al. Completeness and usability of ethnicity data in UK-based primary care and hospital databases. J Public Health (Oxf) 2014; 36:684–92.
- 13 European Commission. Implementation report on the Commission Communication on Rare Diseases: Europe's challenges and Council Recommendation of 8 June 2009 on an action in the field of rare diseases. In: Report from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions. Brussels: European Commission, 2014.
- 14 Petherick ES, Pickett KE, Cullum NA. Can different primary care databases produce comparable estimates of burden of disease: results of a study exploring venous leg ulceration. Fam Pract 2015; 32:374–80.
- 15 Thorslund K, Seifert O, Nilzen K et al. Incidence of bullous pemphigoid in Sweden 2005–2012: a nationwide populationbased cohort study of 3761 patients. Arch Dermatol Res 2017; 309:721–7.
- 16 Nanda A, Dvorak R, Al-Saeed K et al. Spectrum of autoimmune bullous diseases in Kuwait. Int J Dermatol 2004; 43:876–81.
- 17 Jung M, Kippes W, Messer G et al. Increased risk of bullous pemphigoid in male and very old patients: a population-based study on incidence. J Am Acad Dermatol 1999; 41:266–8.
- 18 Serwin AB, Musialkowska E, Piascik M. Incidence and mortality of bullous pemphigoid in north-east Poland (Podlaskie Province), 1999–2012: a retrospective bicentric cohort study. Int J Dermatol 2014; 53:e432–7.
- 19 Wong SN, Chua SH. Spectrum of subepidermal immunobullous disorders seen at the National Skin Centre, Singapore: a 2-year review. Br J Dermatol 2002; 147:476–80.
- 20 Kridin K, Bergman R. Ethnic variations in the epidemiology of bullous pemphigoid in Israel. Int J Dermatol 2018; 57:34–9.
- 21 Marazza G, Pham HC, Scharer L et al. Incidence of bullous pemphigoid and pemphigus in Switzerland: a 2-year prospective study. Br J Dermatol 2009; 161:861–8.
- 22 Bertram F, Bröcker EB, Zillikens D, Schmidt E. Prospective analysis of the incidence of autoimmune bullous disorders in Lower Franconia, Germany. J Dtsch Dermatol Ges 2009; 7:434–40 [Article in German].
- 23 Gudi VS, White MI, Cruickshank N et al. Annual incidence and mortality of bullous pemphigoid in the Grampian region of North-east Scotland. Br J Dermatol 2005; 153:424–7.
- 24 Försti AK, Jokelainen J, Timonen M, Tasanen K. Increasing incidence of bullous pemphigoid in Northern Finland: a retrospective database study in Oulu University Hospital. Br J Dermatol 2014; 171:1223–6.
- 25 Joly P, Baricault S, Sparsa A et al. Incidence and mortality of bullous pemphigoid in France. J Invest Dermatol 2012; 132:1998– 2004.
- 26 Cordel N, Renier M, Samyn A et al.; la Société de dermatologie de la Guadeloupe. [Epidemiology of bullous pemphigoid in Guadeloupe (French West Indies)]. Ann Dermatol Venereol 2009; 136:907–9 [Article in French].
- 27 Brick KE, Weaver CH, Lohse CM et al. Incidence of bullous pemphigoid and mortality of patients with bullous pemphigoid in Olmsted County, Minnesota, 1960 through 2009. J Am Acad Dermatol 2014; 71:92–9.

- 28 Wertenteil S, Garg A, Strunk A et al. Prevalence estimates for pemphigoid in the United States: a sex-adjusted and age-adjusted population analysis. J Am Acad Dermatol 2019; 80:655–9.
- 29 Kridin K, Bergman R. Mortality in patients with bullous pemphigoid: a retrospective cohort study, systematic review and metaanalysis. Acta Dermato-Venereol 2019; 99:72–7.
- 30 Kuan V, Denaxas S, Gonzalez-Izquierdo A et al. A chronological map of 308 physical and mental health conditions from 4 million individuals in the English National Health Service. Lancet Digit Health 2019; 1:e63–77.
- 31 Langan SM, Groves RW, West J. The relationship between neurological disease and bullous pemphigoid: a population-based casecontrol study. J Invest Dermatol 2011; 131:631–6.
- 32 Douros A, Rouette J, Yin H et al. Dipeptidyl peptidase-4 inhibitors and the risk of bullous pemphigoid among patients with type 2 diabetes. Diabetes Care 2019; **42**:1496–503.
- 33 Lee SG, Lee HJ, Yoon MS et al. Association of dipeptidyl peptidase 4 inhibitor use with risk of bullous pemphigoid in patients with diabetes. JAMA Dermatol 2019; 155:172–7.
- 34 Temel AB, Murrell DF. Diagnostic criteria and phenotypes of pemphigoid and the association with gliptins. JAMA Dermatol 2019; 155:147–8.
- 35 Varpuluoma O, Försti AK, Jokelainen J et al. Vildagliptin significantly increases the risk of bullous pemphigoid: a Finnish nationwide registry study. J Invest Dermatol 2018; 138:1659–61.
- 36 Tasanen K, Varpuluoma O, Nishie W. Dipeptidyl peptidase-4 inhibitor-associated bullous pemphigoid. Front Immunol 2019; 10:1238.
- 37 Curtis HJ, Dennis JM, Shields BM et al. Time trends and geographical variation in prescribing of drugs for diabetes in England from 1998 to 2017. Diabetes Obes Metab 2018; **20**:2159–68.
- 38 Wilkinson S, Douglas I, Stirnadel-Farrant H et al. Changing use of antidiabetic drugs in the UK: trends in prescribing 2000–2017. BMJ Open 2018; 8:e022768.
- 39 Public Health England. A Review of Recent Trends in Mortality in England, GW-686. London: Public Health England, 2018.

Supporting Information

Additional Supporting Information may be found in the online version of this article at the publisher's website:

Table S1 Univariate and multivariate analysis of incidenceof bullous pemphigoid in UK primary care per 100 000 per-son years.

Table S2 HR for death, stratified by matched set, comparing patients with bullous pemphigoid and matched disease-free people from general practices in the UK.

Table S3 Published studies reporting the incidence orprevalence of bullous pemphigoid.

Figure S1. Age-adjusted incidence rate of bullous pemphigoid in English Hospital Episode Statistics-linked general practices and hospital inpatient records.

Figure S2. Point prevalence of bullous pemphigoid on 31 July of each calendar year in all of the CPRD, per 100 000 people.

Figure S3. Kaplan–Meier curve showing mortality over time since diagnosis of bullous pemphigoid or index date in disease-free individuals in all CPRD practices.

Text S1. Details of matching individuals with and without bullous pemphigoid for mortality analyses.

Text S2. Incidence sensitivity analysis.

Text S3. Prevalence sensitivity analysis.

Text S4. Mortality sensitivity analysis.

Powerpoint S1 Journal Club Slide Set.

Video S1 Author video.